Author's response to reviews

Title: Ganglioneuroma; A rare cause of adrenal incidentaloma: a case report

Authors:

mine adas (mineadas@yahoo.com)
bora koc (drborakoc@hotmail.com)
gokhan adas (gokhantolgaadas@gmail.com)
filiz ozulker (begumsirins@hotmail.com)

Version: 6 Date: 3 February 2014

Author's response to reviews: see over
Author’s response to reviews

Title: Ganglioneuroma; a rare cause of adrenal incidentaloma: a case report

Authors:
Mine Adas, Bora Koc, Gokhan Adas, Filiz Ozulker

Version: 3   Date: 27-01-2014

Author’s response to reviews: see over

Response to Editor:

1- This case has already controlled for copy-editing for the English Grammar.
2- The article improved by adding the following reference and including some information from it (regarding hormone secretion: Endocr Relat Cancer. 2003 Mar;10(1):99-107. (yellow printed)
3- We include the Ethnicity of the patient in the Abstract and Case presentation sections.
4- We reposition the figure title and legend section after the reference list.
5- We included the appropriate headers for each section.